Author’s response to reviews

Title: Definitions, measurements and prevalence of Fear of Childbirth: A systematic review

Authors:

Christina Nilsson (christina.nilsson@hb.se)
Eva Hessman (eva.hessman@ub.gu.se)
Helen Sjöblom (helen.sjoblom@ub.gu.se)
Anna Dencker (anna.dencker@gu.se)
Elisabeth Jangsten (elisabeth.jangsten@gu.se)
Margareta Mollberg (margareta.mollberg@gu.se)
Harshida Patel (harshida.patel@gu.se)
Carina Sparud-Lundin (carina.s-lundin@fhs.gu.se)
Helena Wigert (helena.wigert@gu.se)
Cecily Begley (cbegley@tcd.ie)

Version: 3 Date: 22 Dec 2017

Author’s response to reviews:

Many thanks for your recent revision. Before editorial acceptance, please address the following editorial requests:

Authors’ response: Thank you. We have reviewed the manuscript and changes have been made according to the BMC guidelines and the suggestions from Editor (point 1 to 7).

1) General formatting: please adhere to our guidelines, here: https://bmcpregnancychildbirth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article.

Answer: The manuscript is revised according to the guidelines

In particular:

* Abstract: this should include four sections, titled:
Background: the context and purpose of the study

Methods: how the study was performed and statistical tests used

Results: the main findings

Conclusions: brief summary and potential implications

Answer: The changes have been made

* Main manuscript: this should include the following main sections, titled:

  Background
  Methods
  Results
  Discussion
  Conclusions

Answer: The changes have been made

Please rename your sections accordingly, and eliminate Objectives in Abstract and Recommendations in the main text. The text included here should be moved under the most relevant section.

Answer: This have been made

2) Declarations: Please include a "Not applicable" statement where relevant (e.g. Acknowledgments and Authors' information)

Answer: Done

3) Competing interests: please add here a statement to indicate that CB and CS-L are members of the BMC Pregnancy and Childbirth Editorial Board.

Answer: The statement is added
4) Please rename "Contribution to authorship" as "Authors' Contributions"
Answer: Done

5) Ethics approval and consent to participate: please rephrase to "Not applicable"
Answer: Done

6) Remove the additional files not for publication: cover letter and response to reviewers
Answer: The files are removed

7) At this stage, please submit a single, clean copy of your manuscript, with no highlights or tracked changes.
Answer: Done